Reviewer 5: • Complexity. Indeed, for $d \gg n$, $d^n$ dominates, which suggests (exact) $d_{OT}$ is actually preferable (see Fig 1).

Reviewer 5: • As you suggest, following proof technique of JDOT paper we are able to show a similar adaptation bound - will add